Author's response to reviews

Title: Mindfulness-based intervention for teenagers with cancer: Rationale and protocol development for a randomized, wait-list clinical trial.

Authors:

Catherine Malboeuf-Hurtubise (catherine.malboeuf-hurtubise@umontreal.ca)
Marie Achille (marie.achille@umontreal.ca)
Serge Sultan (serge.sultan@umontreal.ca)
Majorie Vadnais (majorie.vadnais.hsj@ssss.gouv.qc.ca)

Version: 4 Date: 4 February 2013

Author's response to reviews: see over
M. Altman, M. Simes,

Please find enclosed revisions for the manuscript entitled: *Mindfulness-based intervention for teenagers with cancer: Rationale and protocol development for a randomized, wait-list clinical trial*, which I am submitting for exclusive consideration of publication as an article in Trials.

This revised version addresses the requested formatting changes. Thus, the trial registration number has been added after the abstract (ClinicalTrials.gov Identifier: NCT01783418), along with a statement of ethical approval from both institutions involved by the project (i.e. Université de Montréal and Sainte-Justine Mother and Child University hospital center), including respective reference numbers of approval. Please note that both ethical approval certificates were added in the “additional files” section of the first manuscript submission. Finally, a statement has been added to confirm informed consent will be obtained from each recruited participant for this project.

I thank you very much for your consideration of my work. Please address all correspondence concerning this manuscript to me and feel free to correspond with me by e-mail at catherine.malboeuf-hurtubise@umontreal.ca

Sincerely,

Catherine Malboeuf-Hurtubise, B. Sc., B.A.
Ph.D. candidate in clinical psychology
Health Psychology Laboratory, Université de Montréal